Author’s response to reviews

Title: Acute human bocavirus 1 infection in child with life-threatening bilateral bronchiolitis and right-sided pneumonia: a case report

Authors:

Inga Ziemele (inga.ziemele@gmail.com)

Man Xu (man.xu@helsinki.fi)

Anda Vilmane (anda.vilmane@rsu.lv)

Santa Rasa-Dzelzkaleja (santa.rasa-dzelzkaleja@rsu.lv)

Lea Hedman (lea.hedman@helsinki.fi)

Klaus Hedman (klaus.hedman@helsinki.fi)

Maria Söderlund-Venermo (maria.soderlund-venermo@helsinki.fi)

Zaiga Nora-Krukle (zaiga.nora@rsu.lv)

Modra Murovska (modra@latnet.lv)

Dace Gardovska (dace.gardovska@rsu.lv)

Version: 2 Date: 21 May 2019

Author’s response to reviews:

Reviewer #2:

Title: the title is corrected. Now is specified, that bronchiolitis was bilateral as well is corrected for right-sided pneumonia. Corrections according to the diagnosis - bilateral bronchiolitis is also made in the abstract and main text.

Abstract/Case Presentation: case description in the abstract (line 37) is now written in the past tense as suggested.
Case Report: in the main text (line 159-161) we included the main reason and explanation according to the length of hospitalization for patient between day 10 and day 17. It was mainly due to the additional tests done that time because of the poor weight gain.

In the line 162-164 now is explained what was the suspected cause of the episode of the fever on day 17. We also included, that second swab test was negative for RSV and Influenza A and B, however, viral self-limiting upper RTI was suspected and patient was treated only with the supportive therapy.

Without the details, but we included the main reason why the patient still was kept in the hospital for the several days and it was because of the family reasons (line 165-167). The suspected illness and received treatment for the second episode of the fever, cough and wheezing is also corrected in the text (line 168-170).